Author's response to reviews

Title: Determinants of unintended pregnancies in rural Ghana

Authors:

    Sebastian Eliason (sakeliason@yahoo.co.uk)
    Frank Baiden (baidenf@yahoo.co.uk)
    Barbara A Yankey (hellobabs8@yahoo.com)
    Kofi Awusabo-Asare (awusabo_asare@yahoo.co.uk)

Version: 6  Date: 23 July 2014

Author's response to reviews:

    The Editor-In-Chief
    BMC Pregnancy and Childbirth
    UK

    MS: 1103988124117735 :Research article: Determinants of unplanned pregnancies in rural Ghana

Dear EIC,

A revision of the manuscript has been done based on the editorial comments as follows:

1. List of abbreviations have been incorporated in the revised manuscript after the conclusions and recommendations.

2. Additional files have been removed

3. Formatting improved to conform to journal guidelines

Thank you

Dr Eliason